Author's response to reviews

Title: Prevalence and Associated Factors of Cervical Intraepithelial Neoplasia: A Cervical Cancer Screening Program Performed on 728,704 Women in Beijing

Authors:

Lixin Tao (taolixin.2008@163.com)
Xia Li (lixia_new@163.com)
Qi Gao (gaoqi@ccmu.edu.cn)
Lei Pan (panlei_helen@163.com)
Lijuan Wu (wujuan811017@163.com)
Yanxia Luo (lyx100@ccmu.edu.cn)
Wei Wang (wei6014@yahoo.com)
Ziheng Zheng (myszihe@126.com)
Lili Han (hanll2004@126.com)
Xiuhua Guo (guoxiu@ccmu.edu.cn)

Version: 2 Date: 18 April 2013

Author's response to reviews:

We revised the paper in competing interests, author contributions, and acknowledgements part. Moreover, we have highlighted the revised parts.